Author's response to reviews

Title: Trabectedin as second line treatment in metastatic myxoid liposarcoma: Case report

Authors:

Irene Zarcos (pipelinebiomedicalresources@hotmail.com)
Jose M Jurado (josemjurado@hotmail.com)
Josefa Carrillo (josefacarrillo@hotmail.com)

Version: 3 Date: 23 January 2012

Author's response to reviews: see over
RE: SUBMISSION TO *Journal of Medical Case Reports*

Dear Editor,

Please find attached the manuscript titled: "TRAJECTEDIN AS SECOND LINE TREATMENT IN METASTATIC MYXOID LIPOSARCOMA" to be considered for publication in *Journal of Medical Case Reports*. I, as first author, sign the letter on behalf of all authors. All authors have made substantial contributions to all of the following: (1) the conception and design of the study, or acquisition of data, or analysis and interpretation of data, (2) drafting the article or revising it critically for important intellectual content, (3) final approval of the version to be submitted. None of the authors has declared to have any conflict of interest.

This is a relevant study because it is an original case report reporting an unusual presentation of a disease (Criteria number 2) and a new treatment strategy with implications to a broad medical audience including medical, radiation, surgical and OB-GYN (Criteria number 4).

Do not hesitate to contact me if you need further information.

Looking forward to hearing from you soon,

Best regards,

Irene Zarcos, MD
Medical Oncology Department,
Hospital Clínico San Cecilio
C/ Dr. Olóriz 16
18012 Granada, Spain
Tel: + 36 958 023 000
Fax: + 34 958 023 171